

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

García-Sánchez, Isabel-María; Monteiro, Sónia Maria da Silva; Piñeiro Chousa, Juan Ramón; Aibar-Guzmán, Beatriz

Article

Climate change innovation: Does board gender diversity matter?

Journal of Innovation & Knowledge (JIK)

Provided in Cooperation with:

Elsevier

Suggested Citation: García-Sánchez, Isabel-María; Monteiro, Sónia Maria da Silva; Piñeiro Chousa, Juan Ramón; Aibar-Guzmán, Beatriz (2023): Climate change innovation: Does board gender diversity matter?, Journal of Innovation & Knowledge (JIK), ISSN 2444-569X, Elsevier, Amsterdam, Vol. 8, Iss. 3, pp. 1-14, https://doi.org/10.1016/j.jik.2023.100372

This Version is available at: https://hdl.handle.net/10419/327284

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



NC ND https://creativecommons.org/licenses/by-nc-nd/4.0/





Journal of Innovation & Knowledge





Climate change innovation: Does board gender diversity matter?



Isabel-María García-Sánchez^{a,*}, Sónia Monteiro^b, Juan-Ramón Piñeiro-Chousa^c, Beatriz Aibar-Guzmán^d

- ^a IME, Universidad de Salamanca, Campus Miguel de Unamuno, Edificio FES, Salamanca 37007, Spain
- ^b Escola Superior de Gestão, Instituto Politécnico do Cávado e do Ave (IPCA), Campus do IPCA. 4750 810, Barcelos, Portugal
- ^c Facultad de Administración de Empresas, Universidad de Santiago Compostela, Av. Alfonso X el Sabio, s/n 27002, Lugo, Spain
- d Facultad de Ciencias Económicas y Empresariales, Universidad de Santiago Compostela, Av. Burgo, s/n, Santiago Compostela 15782, Spain

ARTICLE INFO

Article History: Received 3 January 2023 Accepted 25 April 2023 Available online 28 April 2023

Keywords:
Climate change
Environmental innovation
Board gender diversity
Human capital
Corporate governance
Critical mass

JEL codes:

M14

044

Q01

ABSTRACT

This paper responds to recent calls to study the role that board gender diversity plays in shaping corporate responses to climate change. It examines whether female board representation is associated with greater climate change innovation and whether its effect is determined by the presence of a greater number of independent female directors or the existence of a critical mass of female directors. The results obtained for a panel data of 3,928 companies over the period 2010–2020 (35,199 observations) confirm that companies with a greater female board representation (both in number and in percentage) are more proactive in terms of investments in climate change innovation. This positive effect does not require a critical mass of female directors and diminishes when the percentage of female directors is higher than 46.78%, a proportion associated with the presence of five or more female directors. Furthermore, female directors influence climate change innovation mainly through their involvement in management as executive directors, rather than through the monitoring and advisory roles that characterize independent directors. The theoretical and practical implications derived from this research are robust to different considerations and methodological approaches.

© 2023 The Authors. Published by Elsevier España, S.L.U. on behalf of Journal of Innovation & Knowledge. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/)

Introduction

The need to fight against climate change and its devastating effects on the planet represents one of the greatest challenges for humanity (IPCC, 2007). As a major emitter of greenhouse gases and waste generation, companies are urged by the various economic, political and social players to take measures to reduce their carbon footprint (Ben-Amar et al., 2017; Caby et al., 2022; Gull et al., 2022). One way to respond to these growing demands is to implement innovative solutions aimed at reducing pollutant emissions and waste generation, as well as resource consumption (García-Sánchez et al., 2020; Haque, 2017). However, such initiatives are costly and can take time to have an effect, so their development requires strong commitment and support from the board (Atif et al., 2021; Gull et al., 2022; He & Jiang, 2019), as the main body responsible for corporate strategies and risk management (Issa & Bensalem, 2022; Nadeem et al., 2020).

The board of directors has a significant influence on a company's sustainability agenda (Haque & Jones, 2020; UNGC, 2012). Certain attributes related to the composition of the board of directors may be conducive to its orientation towards sustainable development and eco-innovation (de Villiers et al., 2011; Galia, 2015; García-Sánchez et al., 2021; Haque, 2017). Specifically, female representation on the board is considered a key driver of corporate sustainability (Ben-Amar et al., 2017; Boulouta, 2013; Shaukat et al., 2016) in terms of renewable energy consumption (Atif et al., 2021), waste management (Gull et al., 2022), greenhouse gas (GHG) emissions (Konadu et al., 2022) and eco-innovation (Issa & Bensalem, 2022; Nadeem et al., 2020). As directors, women are more mindful of corporate social responsibilities (Issa & Bensalem, 2022) and environmental concerns (Liu, 2018). In addition, they are more open to adopting new ideas and innovative solutions (Konadu et al., 2022). This suggests that female directors are more inclined to support environmental protection initiatives to mitigate climate change (Atif et al., 2021; He & Jiang, 2019) and, therefore, firms with a higher female representation on the board would be more prone to invest in climate change mitigation innovations and technologies.

^{*} Corresponding author. E-mail address: lajefa@usal.es (I.-M. García-Sánchez).

Previous research on the role of gender diversity on boards in driving companies to reduce the impact of their activities on global warming by investing in climate change mitigation innovations and technologies is scarce (He & Jiang, 2019; Le Loarne-Lemaire et al., 2021: Nadeem et al., 2020) and tends to focus on corporate social responsibility (CSR) practices generally (Ben-Amar et al., 2017) or on specific dimensions of corporate environmental sustainability, such as energy consumption, waste management and carbon performance (Atif et al., 2021; Gull et al., 2022; Konadu et al., 2022). However, climate change is a "multi-dimensional concept" and, therefore, business efforts to tackle it must be viewed from a wide perspective. Such perspective should consider business investment in innovations, technologies and other climate change initiatives that are developed at all stages of business activity, with a focus on the organization as a whole. Furthermore, researchers have tended to assume that female directors have "a homogeneous effect" instead of analysing "when" female representation on the board has significant influence on firm decisions (Moreno-Ureba et al., 2022). As a result, prior studies document contradictory results regarding the effect of female directors on eco-innovation (Issa & Bensalem, 2022; Naveed et al., 2022; Zaman et al., 2022).

This paper aims to fill these gaps in the literature and respond to recent calls to delve into the role that board gender diversity plays in shaping corporate responses to climate change (Griffin et al., 2021; Shui et al., 2022; Le Loarne-Lemaire et al., 2021; Baker et al., 2020; Liao et al., 2015). Specifically, we investigate the effect that female representation on the board has on the adoption of different innovations aimed at fighting climate change by attempting to answer two main research questions: (1) is board gender diversity associated with greater climate change innovation? and (2) is the effect of board gender diversity on climate change innovation determined by the presence of a greater number of independent female directors or by the existence of a critical mass of female directors?

To answer these questions, we analyse a panel data of 3928 companies over the period 2010-2020 (35,199 observations) and consider a broad set of investments in different innovations and technologies of different nature, such as management systems, green design, eco-efficiency, pollution control or low-carbon energy (Wang et al., 2018), which help companies to reduce emissions and mitigate the consequences of climate change at all stages of business activity. The results confirm that companies with a greater presence of women on their boards of directors are more proactive in terms of investments in innovation for climate change mitigation. However, this positive relationship diminishes when boards begin to balance in favour of the female gender, with an inverse relationship between board gender diversity and climate change mitigation innovation being observed when the percentage of board seats held by women is higher than 46.78%, a proportion associated with the presence of five or more female board members. Furthermore, the channel through which female directors influence climate change innovation is mainly their involvement in management as executive directors, rather than through the monitoring and advisory roles that characterize independent directors. The theoretical and practical implications derived from this research are robust to different considerations and methodological approaches.

This study extends existing research on the relationship between board gender diversity and corporate environmental proactivity by analysing the impact of female directors on climate change mitigation innovations and technologies, thereby providing a new perspective that has not been previously investigated. Thus, our findings contribute to understanding the dynamics of board gender diversity and its effect on corporate responses to climate change by providing a broad picture of whether, how and under what conditions female board representation influences climate change innovation. Furthermore, this study considers the existence of a non-linear relationship between board gender diversity and corporate sustainability, an aspect little analysed in the literature, thus contributing to fill a

research gap and providing valuable evidence that could explain mixed prior results.

The rest of the paper is organized as follows: Section 2 presents the theoretical framework; Section 3 develops the research hypotheses; Section 4 outlines the research methodology; and Section 5 presents the main findings, which are supplemented in Section 6 with a complementary analysis. Section 7 discusses the study's findings and explores their theoretical and practical implications. Finally, Section 8 draws the main conclusions of the research and presents its limitations and some avenues for future research.

Theoretical framework

Several theories, such as agency theory, resource dependence theory, upper echelons theory, stakeholder theory, legitimacy theory, gender socialization theory, gender role theory, and critical mass theory, have been employed to analyse the relationship between board gender diversity and firms' environmental proactivity. Since each single theory has limited ability to explain the effect of the presence of women on the board of directors on corporate environmental proactivity (Haque & Jones, 2020; Issa & Bensalem, 2022; Konadu et al., 2022; Moreno-Ureba et al., 2022), most authors employ a multi-theoretical framework that combines different theoretical lenses to rationalize this association. In line with this overall approach, we rely on resource dependence theory, gender socialization theory, upper echelons theory, agency theory and critical mass theory to explain the effect of female representation on the board on the adoption of climate change mitigation innovations.

Resource dependence theory (Pfefer & Salancik, 1978) is related to the board's advisory and resource provisioning role, while agency theory is associated with the monitoring role of the board of directors (Ben-Amar et al., 2017; de Villiers et al., 2011; Haque, 2017; Haque & Jones, 2020). From the first perspective, the board of directors provides firms unique resources and capabilities, such as strategic advice, reputation, knowledge, financial resources and networks, which affect the firm's ability and propensity to eco-innovate (Haque, 2017). Specifically, female directors bring different human and relational capital to the board (Caby et al., 2022; Haque & Jones, 2020), expanding its know-how, creativity, competence, expertise, openmindedness and connections (Konadu et al., 2022).

On the other hand, agency theory explains the oversight role that the board of directors plays in constraining management's opportunistic behaviour (Fama & Jensen, 1983) and resolving agency problems arising from the uncertain and long-term nature of investments in climate change mitigation innovations (Griffin et al., 2019; Haque, 2017). From this point of view, the effectiveness of the board of directors in performing its monitoring role is largely influenced by certain attributes, such as the independence of its members (de Villiers et al., 2011; Moreno-Ureba et al., 2022). As for female directors, previous studies have shown that they improve board governance (Seebeck & Vetter, 2022) through effective oversight (Atif et al., 2021; Gull et al., 2022; Kassinis et al., 2016; Konadu et al., 2022) and greater independence (Adams & Ferreira, 2009; Carter et al., 2003).

Upper echelons theory (Hambrick & Mason, 1984) assumes that directors' demographic characteristics and experiences shape their values and behaviours, which in turn influence their decisions and management styles (Hambrick, 2007). From this perspective, gender brings unique attributes to directors (Khandelwal et al., 2020), affecting their attitude, priorities and motivations, which in turn reflect on corporate strategies and decision-making. In this sense, female directors' distinctive backgrounds increase the board's cognitive diversity, making it more likely to accept and promote environmental innovations (Galia et al., 2015; Konadu et al., 2022).

According to gender socialization theory, women and men have different attitudes and perspectives regarding environmental issues as a result of their education (Eagly & Johannesen-Schmidt, 2001;

Konadu et al., 2022; Liu, 2018). Women have been educated to care about others (Moreno-Ureba et al., 2022), which makes them more sensitive to environmental and social challenges (Ben-Amar et al., 2017; Caby et al., 2022; Issa & Bensalem, 2022). From this perspective, female directors will behave in line with these internalized attributes (Atif et al., 2021; Gull et al., 2022; Nadeem et al., 2020) and, therefore, their presence on the board will favour a greater stakeholder and long-term orientation (Issa & Bensalem, 2021), thereby promoting climate change mitigation initiatives.

Hypothesis development

Board gender diversity and climate change mitigation innovation

Prior research has documented an overall positive effect of board gender diversity on firm innovation (Chen et al., 2018; Galia & Zenou, 2012; Le Loarne-Lemaire et al., 2021; Midavaine et al., 2016; Torchia et al., 2011). These findings are attributed to the fact that female directors bring new perspectives, skills and expertise to the board, which fosters creativity and innovative thinking (Galia et al., 2015; Issa & Bensalem, 2022; Nadeem et al., 2020; Naveed et al., 2022). Likewise, prior studies reveal a general positive relationship between board gender diversity and CSR, in general, and environmental proactivity, in particular (Lu & Herremans, 2019; McGuiness et al., 2017; Nielsen & Huse, 2010; Pan et al., 2020). This positive effect is explained because, compared to male directors, female directors are more sensitive to environmental concerns and more environmentally proactive (Haque & Jones, 2020; He & Jiang, 2019; Liao et al., 2015; Nadeem et al., 2020; Pan et al., 2020; Post & Byron, 2015) and tend to show a better awareness of environmental initiatives (Arayssi et al., 2016).

Several authors provide empirical evidence of an overall positive association between board gender diversity and eco-innovation and climate change mitigation initiatives. Regarding eco-innovation, studies such as Galia et al. (2015), Glass et al. (2016), Horbach and Jacob (2018), He and Jiang (2019), Liao et al. (2019), Moreno-Ureba et al. (2022), Nadeem et al. (2020) and Naveed et al. (2022) document a positive impact of the presence of female directors on the board on eco-innovation. A similar relationship has been observed with respect to climate change mitigation initiatives. For example, Atif et al. (2021) showed a positive effect of gender diversity on boards on renewable energy consumption and Gull et al. (2022) documented a positive effect on waste management. Haque (2017) found a positive impact of the presence of female directors on the board on the firm's efforts to reduce emissions and increase eco-efficiency, and García Martín and Herrero (2020) and Konadu et al. (2022) found a negative relationship between board gender diversity and carbon emissions. Other authors (e.g. Ararat & Sayedy, 2019; Ben-Amar et al., 2017; Liao et al., 2015; Tingbani et al., 2020) document a positive relationship between board gender diversity and voluntary disclosure of climate change information. Few studies show negative or no impact of board gender diversity on eco-innovation (e.g. García-Sánchez et al., 2021) and carbon performance (Caby et al., 2022) and disclosures (e.g. Prado-Lorenzo & García-Sánchez, 2010).

Taken together, this empirical evidence suggests that gender-diverse boards are more likely to promote the adoption of climate change mitigation innovations. From the resource dependence theory's perspective, this relationship is supported by the fact that female directors' human and relational capital favour the inclusion of climate change issues into corporate strategies (He & Jiang, 2019; Hollindale et al., 2019; Nadeem et al., 2020), thereby positively affecting innovation in climate change mitigation. Likewise, from the agency theory's standpoint, the increased board oversight due to the presence of female directors contributes to solving agency problems associated with the uncertain and long-term nature of investments in climate change mitigation innovations, by curbing opportunistic

management behaviours that would inhibit this type of investment (Haque, 2017).

Upper echelons theory and gender socialization theory also support the existence of a positive effect of board gender diversity on climate change mitigation innovations. According to the former, the presence of female directors on the board broadens the cognitive composition of the board of directors, which favours a better understanding of societal demands and challenges, such as climate change (Galia et al., 2015; Horbach & Jacob, 2018; Konadu et al., 2022), making the board more likely to support investments in climate change mitigation innovations and technologies. From the perspective of gender socialization theory, female directors will behave in line with the attributes instilled in them. Thus, compared to their male counterparts, female directors are more sensitive to environmental concerns (Nadeem et al., 2020) and have higher ethical standards (Moreno-Ureba et al., 2022); consequently, they will promote proactive environmental strategies and investments (Atif et al., 2021; Gull et al., 2022; Issa & Bensalem, 2021; Nadeem et al., 2020).

Based on this discussion, we assume that gender-diverse boards are more likely to promote the adoption of climate change mitigation innovations. Hence, we pose the following hypothesis:

H1: Board gender diversity is associated with greater climate change innovation.

Female independent directors and climate change mitigation innovation

According to the agency theory's tenets, independent directors perform more effective oversight of management with regard to environmental issues (de Villiers et al., 2011; Haque, 2017) and, therefore, greater board independence would constrain potential opportunistic behaviour by management to delay investments in climate change mitigation innovations and technologies (He & Jiang, 2019). Furthermore, from the perspective of the resource dependence theory, independent directors provide expertise advice and a critical view (Haque, 2017) and a higher stakeholder orientation (Moreno-Ureba et al., 2022), which would favour this type of investment. The greater the number of independent directors on the board, the greater the likelihood that the board will promote climate change innovations.

Consequently, it would be expected that the positive effect of female directors on climate change mitigation innovation would be greater when they are also independent directors. Prior empirical evidence supports this conjecture. Gull et al. (2022) found that the presence of female directors on the board has a positive impact on waste management only when they are independent, concluding that it is this attribute which drives the effect of board gender diversity on waste management. Atif et al. (2020) obtained the same result with regard to sustainable investments, while Atif et al. (2021) found that the impact of female directors on renewable energy consumption was higher when they were also independent. Lastly, Liao et al. (2019) document a positive effect of female independent directors on environmental innovations.

Based on this discussion, we assume that the independence of female directors is what drives the effect of board gender diversity on climate change innovation. Hence, the following hypothesis is stated:

H2: The effect of board gender diversity on climate change innovation is driven by the presence of female independent directors.

Critical mass of female directors and climate change mitigation innovation

Although, according to theoretical arguments and empirical evidence, board gender diversity is assumed to have a positive impact on eco-innovation and climate change mitigation initiatives, the literature argues that a certain number of female directors on the board are required to have a significant impact on business decisions and

strategies (Amorelli & García-Sánchez, 2020; Atif et al., 2021; Ben-Amar et al., 2017; Gull et al., 2022; He & Jiang, 2019; Joecks et al., 2013; Konrad et al., 2008; Kramer et al., 2006; Moreno-Ureba et al., 2022). This view is based on critical mass theory (Kanter, 1977), which postulates that the weight of the minority group influences interpersonal interactions and group dynamics, so that it is necessary for the minority group to reach a critical mass to significantly affect group decision-making. Under that threshold, minority group members are seen as "tokens" representing a particular group and tend to adjust their behaviour to the majority group's criteria but, once the minority group reaches a critical mass, they feel more confident and become influential, thereby improving group dynamics and interactions (Atif et al., 2020; Nuber & Velte, 2021; Terjesen et al., 2009).

From this perspective, female directors would constitute the minority group on the board (Seeker & Vetter, 2022), with a critical mass being necessary to have "presence and voice" and, thus, affect board decisions (Kanter, 1977). Results from previous studies support this argument. Torchia et al. (2011) found that three or more female directors are needed to significantly affect firm innovation, and Moreno-Ureba et al. (2022) obtained the same result in the case of environmental innovation, while He and Jiang (2019) set the threshold at two female directors to exert a significant influence on board decisions regarding green innovation. Atif et al. (2020) and Atif et al. (2021) found that female board representation should reach at least two female directors to have a significant influence on sustainable investments and renewable energy consumption, respectively. Gull et al. (2022) showed that gender diversity significantly affects waste management only when there are at least two female directors on the board, and Ben-Amar et al. (2017) and Hollindale et al. (2019) obtained the same result with regard climate change disclosure.

In light of this discussion, the following hypothesis is formulated: H3: The effect of board gender diversity on climate change innovation is driven by the existence of a critical mass of female directors.

Empirical design

Sample

Although the fight against climate change requires the commitment of the entire business fabric, larger companies should be more involved, mainly because they have greater resources and, as a result of their volume of activity, have a greater environmental impact. In this regard, based on the literature review, the companies included in Compustat and Thomson Reuters EIKON databases were selected.

Three search criteria were used to identify the companies to be analysed: firstly, that both databases contained the financial and non-financial (environmental social and governance, ESG) information needed to construct the variables that would allow estimation of the empirical models designed to test the research hypotheses. Secondly, the information should be available from 2010 to 2020, a period in which there is a clear concern about climate change, and greater homogeneity with respect to the commitments and objectives to be achieved in this regard. Thirdly, that the companies had all the information available for at least five of the eleven years analysed to ensure the robustness of the results through different methodological estimates.

The final sample is made up of 3928 companies, forming an unbalanced panel of 35,199 observations. In addition, information on the institutional environment was incorporated from macroeconomic indicators from different agencies.

Empirical models and methodology

To test the research hypotheses, three empirical models have been designed. In Model 1, the verification of hypothesis H1 will require that β 1>0 and significant from the econometric point of

view. The testing of hypothesis H2, concerning the impact of female directors' independence, will be performed by estimating Model 2 and will require that $\propto_1 > 0$ and significant. The checking of hypothesis H3 will be performed by estimating Model 3, expecting that $0 < \delta_1 < \delta_2 < \delta_3$ and that all coefficients are significant.

$$\begin{split} &\text{Inno_ClimaCh}_{i,t} = \beta_0 + \beta_1 \text{BoardDiversity/WomenDirectors}_{i,t} + \sum\limits_{n=2}^{12} \beta_i \\ &F_{i,t} + \sum\limits_{n=13}^{19} \beta_i B_{i,t} + \sum\limits_{n=20}^{23} \beta_i I_{i,t} + \beta_{24} \text{Country}_i + \beta_{25} \text{Ind}_i + \beta_{26} \text{Year}_t + \epsilon_{it} \\ &+ \eta_i \text{ [Model 1]} \end{split}$$

$$\begin{split} &Inno_ClimaCh_{i,t} = -\alpha_0 + -\alpha_1 Independent_WD_{i,t} + \alpha_2 Insider_W\\ &D_{i,t} + \sum\limits_{n=3}^{13} \alpha_i F_{i,t} + \sum\limits_{n=14}^{20} \alpha_i B_{i,t} + \sum\limits_{n=21}^{24} \alpha_i I_{i,t} + \alpha_{25} Country_i + \alpha_{26} Ind_i + \\ &\alpha_{27} Year_t + \epsilon_{it} + \eta_i \text{ [Model 2]} \end{split}$$

$$\begin{split} &Inno_ClimaCh_{i,t} = \delta_0 + \delta_1 Skewed_{i,t} + \delta_2 Tilted_{i,t} + \delta_3 Balanced_{i,t} + \sum_{n=4}^{14} \delta_i \\ &F_{i,t} + \sum_{n=15}^{21} \delta_i B_{i,t} + \sum_{n=22}^{25} \delta_i I_{i,t} + \delta_{26} Country_i + \delta_{27} Ind_i + \delta_{28} Year_t + \epsilon_{it} + \eta_i \text{ [Model 3]} \end{split}$$

Given the ordinal nature of the dependant variable, the models will be estimated with STATA 17.0 software, using ordinal regressions for panel data, random effects. In order to obtain robust results, static (t-estimates) and dynamic models will be used applying one to three lags (t-1, t-2 and t-3) as instruments for the explanatory variables.

Variables

The dependant variable (Inno_ClimaCh) comprises a broad set of investments in different innovations and technologies related to climate change, such as management system technologies, green design technologies, eco-efficiency technologies, pollution control technologies or low-carbon energy technologies (Wang et al., 2018), which help to reduce emissions and mitigate the consequences of climate change, generating greater climate knowledge (PwC, 2021).

Thus, the ClimateTech variable has been configured as a score formed by the sum of seven items that identify business investment in innovations, technologies and other initiatives against climate change. These projects are of different natures and are developed in all stages of business activity, with a focus on the organization as a whole. The score is generated by adding the following dichotomous variables that identify that the company has made innovations in (i) green building, including pollution-control equipment, and (ii) resource uses with the aim of reducing the consumption of resources and energy and improving resource energy efficiency; has invested in (iii) clean technology in production planning and management processes; and (iv) waste technology with the aim of employing recycling techniques and using recycled materials. In addition, the company has developed (v) environmentally responsible products and promoted initiatives that encourage (vi) sustainable land use (land environment), and (vii) sustainable mobility of its personnel in all work-related travels.

The independent variables designed to test the research hypotheses are BoardDiversity and WomenDirectors for hypothesis H1; Independent_WD and Insider_WD for hypothesis H2; and Skewed, Tilted, and Balanced for hypothesis H3. Following Atif et al. (2020, 2021) and Gull et al. (2022), we define gender diversity by BoardDiversity and WomenDirectors, both variables represent the presence of female directors on the board expressed as a percentage of total number of directors and the number of female directors on the board, respectively. Independent_WD and Insider_WD correspond to the proportion of female independent and executive directors with respect to board size, respectively (Atif et al., 2021; Liao et al., 2019). Finally, with respect to hypothesis H3 based on critical mass theory,

following Seebeck and Vetter (2022), we consider the three types of boards defined by Kanter (1977): skewed (male-dominated boards), tilted (boards with a greater female presence, but without reaching a balanced distribution of seats), and balanced (gender-balanced boards). To identify these situations, three dummies have been created: Skewed, Tilted, and Balanced, which take, respectively, the value 1 when the percentage of women on the board is less than or equal to 20%; is between 20% and 40%; or is greater than 40% and less than or equal to 60%, and 0 otherwise.

To mitigate bias in the estimation of the models, three vectors of variables, F, B, and I, have been included with the aim of controlling for the effect that different variables have shown to have in previous studies (e.g. Aibar-Guzmán et al., 2022; García-Sánchez et al., 2020). Specifically, vector F identifies 11 business characteristics associated with the existence of resources, capabilities and incentives or the involvement of certain financial agents, such as financial analysts and institutional investors. Vector B identifies seven dimensions of board effectiveness as an internal corporate governance mechanism (de Villiers et al., 2011; Galia, 2015; García-Sánchez et al., 2021; Haque, 2017). Lastly, vector I entails the inclusion of four macroeconomic indicators that identify the institutional context and the pressures that may exist on environmental issues at the country level, as well as the effect of the COVID-19 pandemic in the year 2020. Table 1 reflects the definition and measurement of the twenty-two control variables included in the models. Additionally, we control for time period, industry and country using the variables Year, Industry, and Country.

Basic and robust results

Descriptive analysis

Table 1 shows the mean and standard deviation of the variables designed for the estimation of empirical models 1 to 3. The mean climate change innovation score (Inno_CliCh) indicates a low level of business commitment in this regard, standing, on average, at 1.58 (standard deviation of 1.13). This means that companies have promoted or invested in 2 or 3 out of the 7 climate change initiatives that make up the score.

In terms of board gender diversity, female directors represent, on average, approximately 16% of board members (BoardDiversity), occupying around 1.59 seats (WomenDirectors). These values are similar to those obtained by Ben-Amar et al. (2017), Gull et al. (2022), and Naveed et al. (2022), and higher than the values reported by Atif et al. (2020) and Atif et al. (2021), but our mean percentage of gender diversity is lower than that obtained by Issa and Bensalem (2022), Konadu et al. (2022) and Nube and Velte (2021). In general, these findings indicate a low participation of women on boards of directors. However, the standard deviation indicates that this percentage can vary significantly amongst the companies analysed.

In addition, 41.9% of companies have only one female director on the board (One_WD), 14.59% of companies have two female directors and 18.38% have three or more female directors. Quantifying the board gender equity, 42.79% of the companies have a skewed board, with a female presence of less than 20%, while 28.05% of the companies have a tilted board, with a percentage of women board members between 20% and 40%. Only 4.11% of the companies have a balanced board.

Regarding the type of female directors, the presence of independent female directors (Independent_WD) is slightly higher than that of executive female directors (Insider_WD). These findings are consistent with the results obtained by Atif et al. (2021) and Gull et al. (2022), who also found that the percentage of independent female directors exceeds the percentage of executive female directors.

Table 2 shows the correlation matrix. As can be seen, the coefficients that determine the bivariate correlation are significant, although their values are very small, indicating the absence of collinearity problems.

Table 1Variables: definition and statistics.

Variable	Definition	Mean	Std. dv.
Inno_CliCh	Score on the degree of innova- tion in the fight against cli- mate change	1.58	1.13
BoardDiversity (BD)	Proportion of female board members to board size	15.67	13.07
WomenDirectors (WD) Independent_WD	Number of female directors Proportion of independent	1.59 8.63	1.45 10.43
insider_WD	female directors Proportion of executive female	7.04	8.76
ogAGE	directors Age of the company as mea- sured by the logarithm of age	3.20	0.90
logTA	Size of the company measured by the logarithm of total assets	15.44	1.82
TobinQ	Ratio of market value to replacement value of the company's assets	0.00	0.00
ROA	Ratio of net income to total assets	2.69	5.13
InterSales	Percentage of the company's sales abroad	25.9	36.2
R&D	Monetary volume of intangi- ble investments	11.3	26.9
Advertising	Monetary volume of advertis- ing expenditure	10.2	10.3
CAPEX	Monetary volume of capital investments	59.8	64.7
Dividend	Annual dividend distributed	28.1	36.1
Leverage Analysts	Corporate indebtedness Number of financial analysts	51.85 13.29	460.84 8.83
InstInve_VR	following the company Voting rights held by institu- tional investors whose	82.88	32.89
	equity stake exceeds 5%		
BoardSize BoardActivity	Number of board members Number of board meetings	10.06 9.56	3.41 5.67
BoardIndep	held in the year Percentage of independent directors in relation to the	48.74	36.85
BoardTenure	size of the board of directors Average length of service of	7.50	3.95
ERRI	board members on the board Environmental Regulatory Regime Index	0.87	0.67
EPI	Environmental Performance Index	58.77	7.20
Skewed	Dummy that takes the value 1 if the percentage of female directors is less than or equal to 20%.	Frequency (%) 42.79	
Tilted	Dummy that takes the value 1 if the percentage of female directors is greater than 20% and less than or equal to 40%.	28.05	
Balanced	Dummy that takes the value 1 if the percentage of female directors is greater than 40% and less than or equal to 60%.	4.11	
One_WD	Dummy that takes the value 1 if there is only one woman on the board	41.98	
Two_WD	Dummy that takes the value 1 if there are two women on the board	14.59	
Three&More_WD	Dummy that takes the value 1 if there are three or more women on the board	18.38	
DStkEng	Dummy that identifies the existence of a policy of active stakeholder participation in management processes (García-Sánchez et al., 2022).	39.32	
CEOduality	Dummy that identifies the duality of functions of the CEO as Chairman of the Board.	62.97	

(continued)

Table 1 (Continued)

Variable	Definition	Mean	Std. dv.
CSRCommittee	Dummy that identifies the existence of a specialized subcommittee on sustain- ability on the boar	50.55	
EU	Dummy that identifies that the company's country of origin is EU for the period 2019 and 2020 (EU) due to the approval of the European Action Plan to ensure sustainable growth	14.19	
Covid	Dummy identifying the effects of the COVID-19 Pandemic in 2020	9.83	

Main results

The results obtained in the estimation of the empirical models designed to test the research hypotheses are summarized in Table 3 (Model 1) and Table 4 (Models 2 and 3). For all models, a preliminary equation, Model 0, is presented, which assumes the estimation of each model in t with the only inclusion of the independent variables relating to gender diversity. That is, in each equation's version 0, the control variables are omitted, except those related to sector, country and time period. In addition, to ensure the robustness of the results obtained in the face of methodological specifications, each model is estimated with and without instruments. This implies the inclusion of the explanatory variables in t (static models or without instruments) and the use of lags t-1, t-2 and t-3 (dynamic models, with lags or with instruments).

Table 3 shows the results relating to Model 1, designed to test hypothesis H1. In this regard, the testing of hypothesis H1 is performed by using the variables BoardDiversity (BD) and WomenDirectors (WD), representing the proportion and number of female directors, respectively. As can be seen, the BoardDiversity variable has a positive impact on the dependant variable Inno_ClimaCh, significant for a 99% confidence level. Specifically, its isolated effect is $\beta_1 = 0.0686$, reducing to $\beta_1 = 0.0243$ when considering other variables representing the resources and capabilities of the companies, the attributes of the board of directors and the existence of institutional pressures. The positive nature of the effect and its significance are maintained in the dynamic models in which lags t-1 ($\beta_1 = 0.0278$), t-2 ($\beta_1 = 0.0310$) and t-3 ($\beta_1 = 0.0417$) are used as instruments.

The effect observed for the WomenDirectors variable is quite similar in sign and significance, although the amount is larger due to the measurement scale. Its effect in Model 0 is β_1 = 0.643, and β_1 = 0.226 in the full static model (equation 1, including control variables). In the dynamic models, the coefficients reach the following values: β_1 = 0.268 in the full model with lag t-1; β_1 = 0.305 in the full model with lag t-2; and β_1 = 0.414) in the full model with lag t-3.

The alternative use of both variables and the similarity of results between static and dynamic models ensures the robustness of the results regarding the role that board gender diversity exerts on corporate commitment to the fight against climate change, identified through the promotion of and investment in different proactive environmental projects and initiatives. Therefore, hypothesis H1, which stated that board gender diversity is associated with greater climate change innovation, can be accepted.

Panel A in Table 4 shows the results corresponding to Model 2, designed to test hypothesis H2. As can be seen, the impact that the presence of women on the board has on the company's proactivity in innovation against climate change is independent of the type of director considered (independent or executive). Thus, the first column of this panel shows that both independent and executive female directors have a positive and significant impact at a confidence level

of 99%. In fact, the magnitude of the impact is greater in the case of the latter type of female directors: the Independent_WD variable has an effect of δ_1 = 0.0169, while the effect of the Insider_WD variable is δ_1 = 0.0333. The nature of the effect, its significance and amount are maintained in the dynamic models estimated with lags t-1 to t-3. Therefore, the evidence obtained from the estimation of Model 2 leads us to reject hypothesis H2, which posited that the effect of board gender diversity on climate change innovation is driven by the presence of independent female directors.

Regarding the last hypothesis, Panel B in Table 4 summarizes the results corresponding to Model 3. These findings do not suggest the theoretically expected relationship in its entirety. It can be observed that the most proactive companies with respect to investing in climate change innovations are those whose boards of directors have a percentage of women equal to or less than 40%. Thus, regarding the static model, in the first column, the Skewed and Tilted variables have a positive impact on the Inno_ClimaCh score, significant at a 99% confidence level. However, the effect of the Tilted variable, representative of a female presence between 20% and 40% of the board size, is greater than that of the Skewed variable, indicative of women occupying at most 20% of the seats on this body $0 < \delta_1 = 0.352 < \delta_2 = 0.479$. In contrast, the Balanced variable, representing boards with a balanced presence of men and women, although having a positive effect, is not significant from an econometric point of view. The results are robust to the use of dynamic models (results showed in the following three columns) and allow us to partially accept hypothesis H3 that the effect of board gender diversity on climate change innovation is driven by the existence of a critical mass of female directors.

In terms of the control variables, the various models confirm that companies with higher levels of resources and capabilities associated with their age and size, as well as those with more effective internal corporate governance structures (i.e. more independent boards and with a higher specialization on sustainability issues) and more active in their dialogue with stakeholders are the most proactive companies in climate change innovation. With regard to national pressures, the European institutional context and the COVID-19 pandemic are drivers of these decisions. As regards the former, our findings reflect the effect of the several reforms introduced in the European Union to encourage business commitment to the fight against climate change (Nube & Velte, 2022). This is also the case for environmental commitment at the country level as measured by the EPI indicator. Conversely, the existence of broader environmental legislation inhibits these proactive practices.

Complementary analysis

The results obtained with respect to critical mass theory, Model 3, suggest the existence of a possible non-linear relationship between board gender diversity and environmental innovation in climate change. To contrast the existence of this relationship and to deepen the understanding of the ideal size of critical mass, the following models have been defined:

$$\begin{split} &\textbf{Inno_ClimaCh/EnvScore/ESGScore}_{i,t} = & 6_0 + & 6_1 \textbf{BoardDiversity/} \\ &\textbf{WomenDirectors}_{i,t} + 6_2 \textbf{Square_BD/SquareWD}_{i,t} + \sum_{n=3}^{13} 6_i F_{i,t} + \sum_{n=14}^{20} 6_i B_{i,t} \\ &+ \sum_{n=21}^{24} 6_i I_{i,t} + 6_{25} \textbf{Country}_i + 6_{26} \textbf{Ind}_i + 6_{27} \textbf{Year}_t + \epsilon_{it} + \eta_i \left[\textbf{Model 4} \right] \end{split}$$

 $Inno_ClimaCh/EnvScore/ESGScore_{i,t} = \Omega_0 + \Omega_1One_WD_{i,t}$

$$\begin{split} &+\Omega_{2}\textbf{Two_WD}_{i,t}+\Omega_{3}\textbf{Three&&}\textbf{More_WD}_{i,t}\\ &+\sum_{n=4}^{14}\Omega_{i}\textbf{F}_{i,t}+\sum_{n=15}^{21}\Omega_{i}\textbf{B}_{i,t}+\sum_{n=22}^{25}\Omega_{i}\textbf{I}_{i,t}+\Omega_{26}\textbf{Country}_{i}\\ &+\Omega_{27}\textbf{Ind}_{i}+\Omega_{28}\textbf{Year}_{t}+\epsilon_{it}+\eta_{i} \end{split} \label{eq:equation:equat$$

Table 2 Bivariate correlations matrix (**** p<0.01, *** p<0.05, * p<0.1).

		1	2	3	4	5	6	7	8	9	10	11	12
1	Inno_CliCh	1											
2	BoardDiversity(BD)	0.19***	1										
3	Skewed	0.14***	0.69***	1									
4	Tilted	0.17***	0.81***	0.40***	1								
5	Balanced	0.10***	0.50***	0.12***	0.30***	1							
6	WomenDirectors(WD)	0.26***	0.89***	0.64***	0.74***	0.43***	1						
7	One_WD	-0.10***	-0.06***	0.49***	-0.25***	-0.17***	-0.17***	1					
8	Two_WD	0.06***	0.24***	0.24***	0.16***	-0.08***	0.19***	-0.35***	1				
9	Three&More_WD	0.22***	0.64***	0.28***	0.61***	0.42***	0.76***	-0.40***	-0.20***	1			
10	Independent_WD	0.18***	0.74***	0.48***	0.62***	0.39***	0.66***	-0.08***	0.17***	0.49***	1		
11	Insider_WD	0.07***	0.61***	0.47***	0.48***	0.29***	0.54***	0.01**	0.16***	0.37***	-0.08***	1	
12	logAGE	0.23***	0.03***	0.01***	0.03***	0.02***	0.09***	-0.06***	0.01***	0.09***	-0.03***	0.07***	1
13	logTA	0.38***	0.07***	0.10***	0.07***	0.01***	0.25***	-0.12***	0.05***	0.22***	0.08***	0.00	0.16***
14	TobinQ	-0.08***	0.03***	0.01***	0.02***	0.00	-0.02***	0.03***	0.01	-0.03***	0.00	0.04***	-0.05***
15	ROA	0.00	-0.06***	-0.08***	-0.04***	-0.01**	-0.06***	-0.03***	-0.02***	-0.03***	-0.04***	-0.04***	0.02***
16	InterSales	0.05***	-0.06***	-0.07***	-0.04***	-0.01***	-0.06***	-0.02***	-0.03***	-0.03***	-0.04***	-0.05***	0.03***
17	R&D	0.03***	-0.03***	-0.04***	-0.03***	-0.01	-0.03***	-0.01	-0.02***	-0.02***	-0.02***	-0.03***	0.02***
18	Advertising	0.04***	-0.08***	-0.08***	-0.06***	-0.02***	-0.07***	-0.01***	-0.04***	-0.04***	-0.06***	-0.05***	0.04***
19	CAPEX	0.05***	-0.07***	-0.08***	-0.05***	-0.02*** -0.02***	-0.07***	-0.01**	-0.03***	-0.04***	-0.05***	-0.05*** -0.05***	0.03***
20	Dividend	0.03***	-0.04***	-0.03***	-0.04***	-0.02 -0.01***	-0.05***	0.01***	-0.03 -0.02***	-0.03***	-0.04***	-0.03 -0.02***	0.03
21	Leverage	0.02***	-0.04	-0.10***	-0.04	-0.02***	-0.09***	-0.03***	-0.02	-0.04***	-0.04	-0.02	0.02
22	Analysts	0.02	0.04***	0.07***	0.04***	0.00	0.15***	-0.03 -0.07***	0.04***	0.13***	0.06***	0.00	0.03
23	InstInve_VR	-0.03***	-0.06***	-0.06***	-0.04*** -0.04***	-0.01**	-0.06***	-0.07 -0.01*	-0.02***	-0.04***	-0.07***	-0.01*	0.04
24	DStkEng	0.53***	0.16***	0.10***	0.14***	0.10***	0.24***	-0.01 -0.10***	0.04***	0.21***	0.17***	0.05***	0.03
25	BoardSize	0.25***	0.10	0.10	0.14	-0.01**	0.24	-0.10 -0.11***	0.04	0.21	0.17	0.03	0.13
26	BoardActivity	0.25	-0.03***	-0.06***	-0.02***	0.03***	-0.04***	-0.02***	-0.04***	-0.01***	-0.01	-0.04***	0.05***
27	CEOduality	0.03	0.01	-0.00 -0.02***	0.02***	0.03	-0.04 -0.02***	0.00	-0.04 -0.02***	0.00	0.03***	-0.04 -0.02***	-0.04***
28		0.04	0.01	0.19***	0.02	0.05	0.19***	0.00	0.07***	0.00	0.62***	-0.02 -0.44***	-0.04 -0.14***
29	BoardIndep	-0.09***	-0.02***	0.19	-0.04***	-0.06***	-0.01**	0.02	0.07	-0.04***	-0.11***	0.10***	0.25***
30	BoardTenure CSRCommittee	-0.09 0.55***	-0.02 0.14***	0.04	-0.04 0.13***	0.07***	-0.01 0.22***	-0.09***	0.04	-0.04 0.19***	-0.11 0.14***	0.10	0.23
31	ERRI	-0.06***	0.14	0.10	0.13	0.07	0.22	0.00	0.05	0.13	0.14	0.04	0.21
32	EPI	-0.06 0.14***	0.24	-0.01*	0.20	0.10	0.20	-0.06***	-0.03***	0.12	0.18	0.13	0.00
32 33	EU	0.14	0.10	-0.01 0.15***	0.10	0.11	0.10	-0.06 -0.10***	-0.03 0.01	0.10	0.06	0.07	0.03
				0.15	0.23								
34	Covid	0.13***	0.17***			0.09***	0.13***	-0.01**	0.04***	0.09***	0.15***	0.07***	0.03***
12	le «TA	13	14	15	16	17	18	19	20	21	22	23	24
13	logTA	1											
14	TobinQ	-0.30***	1	4									
15	ROA	0.01	0.01**	1									
16	InterSales	0.07***	-0.02***	0.10***	1								
17	R&D	0.05***	-0.01	0.07***	0.90***	1							
18	Advertising	0.08***	-0.01	0.10***	0.86***	0.84***	1						
19	CAPEX	0.08***	-0.02***	0.12***	0.75***	0.74***	0.82***	1					
20	Dividend	0.06***	0.01	0.13***	0.29***	0.25***	0.43***	0.52***	1				
21	Leverage	-0.01	0.00	0.16***	0.13***	0.03***	0.30***	0.32***	0.28***	1			
22	Analysts	0.47***	0.05***	0.06***	0.12***	0.09***	0.12***	0.11***	0.05***	0.03***	1		
23	InstInve_VR	-0.05***	0.01**	0.02***	0.02***	0.02***	0.03***	0.03***	0.00	0.01***	-0.03***	1	
24	DStkEng	0.32***	-0.08***	0.02***	0.06***	0.04***	0.05***	0.06***	0.04***	0.02***	0.25***	0.00	1
25	BoardSize	0.51***	-0.13***	-0.01**	0.00	0.00	-0.01	0.01*	-0.03***	-0.06***	0.25***	0.01*	0.23***
26	BoardActivity	0.12***	-0.09***	0.02***	0.01*	0.00	0.02***	0.05***	0.09***	0.01	-0.06***	-0.01***	0.05***
27	CEOduality	-0.04***	-0.02***	0.01	-0.01**	-0.02***	0.00	0.01	0.02***	0.04***	-0.09***	-0.01**	0.10***
28	BoardIndep	0.05***	-0.01***	-0.04***	0.01	0.00	-0.02***	-0.01***	-0.01***	-0.04***	0.06***	-0.06***	0.11***
29	BoardTenure	-0.07***	0.05***	-0.05***	-0.05***	-0.03***	-0.06***	-0.06***	-0.04***	-0.05***	-0.06***	-0.01*	-0.12***
30	CSRCommittee	0.32***	-0.08***	0.01**	0.05***	0.03***	0.04***	0.04***	0.01***	0.01	0.28***	-0.01	0.55***
31	ERRI	-0.11***	0.02***	-0.10***	-0.10***	-0.05***	-0.14***	-0.15***	-0.16***	-0.22***	-0.13***	-0.06***	-0.16***
32	EPI	0.10***	-0.04***	-0.01**	-0.01**	0.00	-0.02***	-0.03***	-0.04***	-0.07***	0.00	0.00	0.02***
33	EU	0.16***	-0.04***	-0.03***	-0.03***	-0.02***	-0.04***	-0.04***	-0.03***	-0.04***	0.16***	0.00	0.21***
34	Covid	-0.02***	0.03***	-0.01***	-0.01	0.00	-0.01	-0.01*	0.00	-0.01*	-0.07***	-0.01	0.09***
		25	26	27	28	29	30	31	32	33	34		
25	BoardSize	1											
26	BoardActivity	0.02***	1										
27	CEOduality	-0.03***	0.10***	1									
28	BoardIndep	-0.01**	-0.02***	0.02***	1								
29	BoardTenure	-0.02***	-0.20***	-0.24***	-0.17***	1							
30	CSRCommittee	0.24***	0.06***	0.06***	0.07***	-0.11***	1						
31	ERRI	-0.11***	-0.10***	-0.08***	0.07	0.12***	-0.03***	1					
32	EPI	-0.11 0.01	-0.10 0.07***	-0.08 0.07***	-0.03***	-0.12 -0.12***	-0.03 0.07***	0.53***	1				
J Z			0.07***	0.07***	0.07***	-0.12 $-0.11***$	0.07	0.53	0.46***	1			
33	HIII												
33 34	EU Covid	0.18*** -0.05***	0.02***	0.07	0.07***	0.02***	0.05***	0.01	-0.01**	0.00	1		

Table 3 Board Diversity and Climate Change innovations (*** p < 0.01, ** p < 0.05, * p < 0.1).

	Mod	el 0 (t)	Mod	el 1 (t)	Lagged M	odel 1 (t-1)	Lagged Mo	odel 1 (t-2)	Lagged Mo	odel 1 (t-3)
	coeff	coeff	coeff	coeff	coeff	coeff	coeff	coeff	coeff	coeff
BoardDiversity(BD)	(std.error) 0.0686*** (0.00155)	(std.error)	(std.error) 0.0243*** (0.00210)	(std.error)	(std.error) 0.0278*** (0.00236)	(std.error)	(std.error) 0.0310*** (0.00269)	(std.error)	(std.error) 0.0417*** (0.00313)	(std.error)
WomenDirectors(WD)		0.643*** (0.0147)		0.226*** (0.0212)		0.268*** (0.0237)		0.305*** (0.0267)		0.414*** (0.0307)
logAGE			0.896*** (0.0540)	0.910*** (0.0541)	0.870*** (0.0571)	0.880*** (0.0571)	0.800*** (0.0604)	0.806*** (0.0604)	0.804*** (0.0654)	0.811*** (0.0654)
logTA			0.878*** (0.0364)	0.875*** (0.0365)	0.971*** (0.0396)	0.966*** (0.0397)	1.066*** (0.0430)	1.061*** (0.0430)	0.969*** (0.0465)	0.965*** (0.0466)
TobinQ			9.123 (14.62)	9.428 (14.61)	-5.851 (16.24)	-6.289 (16.24)	56.35*** (17.09)	55.56*** (17.08)	49.58*** (18.70)	48.67*** (18.72)
ROA			-0.000512 (0.000573)	-0.000522 (0.000573)	-0.000361 (0.000588)	-0.000375 (0.000587)	-0.000274 (0.000617)	-0.000287 (0.000616)	-0.000150 (0.000654)	-0.000160 (0.000654)
InterSales			0.000 (0.000)	0.000 (0.000)	0.000 (0.000)	0.000 (0.000)	5.93e-11** (0.000)	5.82e-11** (0.000)	0.0001* (0.000)	0.000 (0.000)
R&D			2.31e-10 (2.28e-10)	2.29e-10 (2.28e-10)	2.74e-10 (2.62e-10)	2.69e-10 (2.62e-10)	1.30e-10 (3.28e-10)	1.17e-10 (3.28e-10)	2.00e-10 (3.97e-10)	1.75e-10 (3.98e-10)
Advertising			-9.54e-11 (1.13e-10)	-9.52e-11 (1.12e-10)	-1.44e-10 (1.18e-10)	-1.42e-10 (1.18e-10)	-1.63e-10 (1.24e-10)	-1.58e-10 (1.24e-10)	-9.05e-11 (1.42e-10)	-8.32e-11 (1.42e-10)
CAPEX			(0.000) (8.02e-11)	(0.000) (8.01e-11)	(0.000) (8.56e-11)	(0.000) (8.56e-11)	(0.000) (9.18e-11)	(0.000) (9.18e-11)	-1.36e-10 (9.54e-11)	-1.25e-10 (9.55e-11)
Dividend			(0.000) (1.35e-10)	(0.000) (1.35e-10)	(0.000) (1.34e-10)	(0.000) (1.35e-10)	-1.43e-10 (1.56e-10)	-1.40e-10 (1.56e-10)	-1.03e-10 (1.73e-10)	-1.01e-10 (1.73e-10)
Leverage			7.96e-05 (0.000336)	8.42e-05 (0.000336)	-7.47e-05 (0.000348)	-7.59e-05 (0.000348)	-5.31e-05 (0.000369)	-5.23e-05 (0.000369)	0.000888** (0.000408)	0.000889** (0.000408)
Analysts			0.0110** (0.00469)	0.0108** (0.00470)	0.0102** (0.00506)	0.0102** (0.00506)	0.00633 (0.00550)	0.00648 (0.00550)	0.0114* (0.00607)	0.0112* (0.00607)
InstInve_VR			-0.00243 (0.00159)	-0.00252 (0.00159)	-0.00382** (0.00168)	-0.00390** (0.00169)	-0.00469*** (0.00178)	-0.00471*** (0.00178)	-0.00340* (0.00192)	-0.00333* (0.00192)
DStkEng			1.227*** (0.0569)	1.232*** (0.0569)	1.036*** (0.0633)	1.036*** (0.0634)	0.827*** (0.0715)	0.823*** (0.0715)	0.747*** (0.0812)	0.740*** (0.0812)
BoardSize			-0.00645 (0.0108)	-0.0356*** (0.0111)	0.00589 (0.0117)	-0.0265** (0.0120)	0.00807 (0.0128)	-0.0272** (0.0131)	0.00376 (0.0142)	-0.0419*** (0.0146)
BoardActivity			-0.0106** (0.00484)	-0.0101** (0.00484)	-0.0245*** (0.00577)	-0.0240*** (0.00577)	-0.0283*** (0.00635)	-0.0278*** (0.00635)	-0.0254*** (0.00736)	-0.0251*** (0.00736)
CEOduality			0.0822 (0.0559) 0.00310***	0.0815 (0.0559)	0.163*** (0.0604)	0.162*** (0.0604)	0.224*** (0.0660)	0.224*** (0.0660)	0.325*** (0.0727) 0.00305***	0.324*** (0.0728)
BoardIndep BoardTenure			(0.00310 ⁻¹ (0.000654) 0.0155	0.00313*** (0.000654) 0.0125	0.00296*** (0.000715) 0.0117	0.00296*** (0.000715) 0.00882	0.00263*** (0.000794) 0.00615	0.00260*** (0.000794) 0.00348	(0.00305** (0.000891) 0.00479	0.00302*** (0.000892) 0.00194
CSRCommittee			(0.00976) 1.683***	(0.00974) 1.684***	(0.0117 (0.0107) 1.151***	(0.0107) 1.149***	(0.0118) 0.805***	(0.0118) 0.801***	(0.0131) 0.500***	(0.0131) 0.494***
ERRI			(0.0599) -0.943***	(0.0598) -0.941***	(0.0664) -1.052***	(0.0664) -1.058***	(0.0743) -1.084***	(0.0743) -1.096***	(0.0832) -1.130***	(0.0833) -1.151***
EPI			(0.132) 0.0948***	(0.132) 0.0953***	(0.140) 0.0966***	(0.140) 0.0976***	(0.146) 0.0912***	(0.147) 0.0926***	(0.157) 0.0897***	(0.157) 0.0917***
EU			(0.0111) 0.831***	(0.0112) 0.808***	(0.0117) 0.874***	(0.0118) 0.841***	(0.0123) 0.864***	(0.0123) 0.823***	(0.0130) 0.878***	(0.0131) 0.810***
Covid			(0.203) 1.819***	(0.204) 1.838***	(0.215) 1.930***	(0.216) 1.948***	(0.226) 2.032***	(0.227) 2.047***	(0.244) 1.919***	(0.245) 1.926***
Industry	yes	yes	(0.0579) yes	(0.0578) yes	(0.0586) yes	(0.0585) yes	(0.0604) yes	(0.0603) yes	(0.0632) yes	(0.0632) yes
Country Year	yes	yes	yes	yes	yes yes	yes yes	yes	yes	yes yes	yes yes
Log Likelihood	yes -38,534.52	yes -38,564.33	yes 21,790.31	yes -21,800.08	yes -19,520.32	yes -19,525.70	yes 17,092.93	yes 17,094.45	yes -14,555.20	yes -14,553.60
p-value	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000

Model 4 has been designed to test for the existence of a possible non-linear relationship between board gender diversity and environmental proactivity in the fight against climate change. For this purpose, both the variables BoardDiversity and WomenDirectors and their squares (Square_BD and Square_WD) have been included. In case of the existence of an inverse-U relationship, it is expected that $6_1 > 0$ $\gamma 6_2 < 0$.

In Model 5, following Atif et al. (2021), we have included the dummies One_WD, Two_WD and Three&More_WD that take the value 1 to identify those companies whose boards have, respectively, 1, 2 and 3 or more female directors. Otherwise, the variables take the value 0. It is expected that $0<\Omega_1<\Omega_2<\Omega_3$.

In addition, in order to show whether this effect is specific to the decisions made with respect to climate change innovation or is

generalizable to all environmental and sustainability areas, the Env-Score and ESGScore variables, constructed from the performance scores available in EIKON for both dimensions, will be used. These models will be estimated using regressions adapted to the nature of the dependant variables, ordinal and Tobit for panel data, controlling for multicollinearity introduced by the squaring of the variables by using centred variables.

The results depicted in Panel A in Table 5 allow us to observe a partial effect of critical mass in the sense that the presence of 3 or more female directors on the board has a greater impact on climate change innovation than if there are one or two female directors, but the presence of a single female director guarantees business proactivity in the fight against climate change (0 $<\Omega_1=0.531<\Omega_2=0.876<$ $<\Omega_3=0.997$).

Table 4 Female director's typology and critical mass (*** p < 0.01, ** p < 0.05, * p < 0.1).

9

				Panel A. Female I	Directors typolog	gy		
	Model 2 (t)	Lagged Model 2 (t-1)	Lagged Model 2 (t-2)	Lagged Model 2 (t-3)	Model 3 (t)	Lagged Model 3 (t-1)	Lagged Model 3 (t-2)	Lagged Model 3 (t-3)
	coeff	coeff	coeff	coeff	coeff	coeff	coeff	coeff
	(std.error)	(std.error)	(std.error)	(std.error)	(std.error)	(std.error)	(std.error)	(std.error)
Independent_WD	0.0169***	0.0197***	0.0208***	0.0315***				
	(0.00287)	(0.00324)	(0.00375)	(0.00441)				
Insider_WD	0.0333***	0.0375***	0.0426***	0.0526***				
	(0.00319)	(0.00354)	(0.00399)	(0.00457)				
Skewed					0.352***	0.382***	0.370***	0.459***
					(0.0469)	(0.0523)	(0.0592)	(0.0681)
Tilted					0.479***	0.515***	0.599***	0.817***
					(0.0573)	(0.0617)	(0.0680)	(0.0762)
Balanced					0.00648	0.0726	0.100	0.116
					(0.0945)	(0.113)	(0.137)	(0.168)
logAGE	0.903***	0.878***	0.808***	0.812***	0.912***	0.886***	0.814***	0.820***
	(0.0541)	(0.0571)	(0.0605)	(0.0655)	(0.0540)	(0.0571)	(0.0604)	(0.0654)
logTA	0.877***	0.971***	1.066***	0.970***	0.881***	0.975***	1.068***	0.969***
	(0.0364)	(0.0396)	(0.0430)	(0.0466)	(0.0365)	(0.0397)	(0.0430)	(0.0465)
TobinQ	8.068	-7.235	55.15***	48.72***	9.146	-5.056	58.22***	51.47***
	(14.61)	(16.22)	(17.14)	(18.73)	(14.66)	(16.32)	(17.05)	(18.70)
ROA	-0.000518	-0.000372	-0.000294	-0.000172	-0.000483	-0.000350	-0.000270	-0.000146
	(0.000572)	(0.000588)	(0.000618)	(0.000657)	(0.000572)	(0.000588)	(0.000618)	(0.000655)
InterSales	0.000	0.000	5.91e-11**	0.0001*	0.000	0.000	6.03e-11**	5.00e-11*
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
R&D	2.41e-10	2.82e-10	1.42e-10	2.18e-10	2.35e-10	2.75e-10	1.22e-10	2.12e-10
	(2.29e-10)	(2.62e-10)	(3.28e-10)	(3.97e-10)	(2.28e-10)	(2.62e-10)	(3.27e-10)	(3.96e-10)
Advertising	-9.92e-11	-1.48e-10	-1.67e-10	-9.40e-11	-9.85e-11	-1.47e-10	-1.65e-10	-9.69e-11
	(1.13e-10)	(1.18e-10)	(1.24e-10)	(1.42e-10)	(1.12e-10)	(1.18e-10)	(1.24e-10)	(1.42e-10)
CAPEX	0.000	0.000	0.000	-1.42e-10	0.000	0.000	0.000	-1.35e-10
	(8.01e-11)	(8.56e-11)	(9.18e-11)	(9.55e-11)	(8.03e-11)	(8.59e-11)	(9.20e-11)	(9.54e-11)
Dividend	0.000	0.000	-1.52e-10	-1.08e-10	0.000	0.000	-1.56e-10	-1.09e-10
	(1.36e-10)	(1.34e-10)	(1.57e-10)	(1.73e-10)	(1.36e-10)	(1.35e-10)	(1.57e-10)	(1.73e-10)
Leverage	0.000106	-5.35e-05	-2.85e-05	0.000915**	8.94e-05	-7.51e-05	-5.66e-05	0.000899**
	(0.000336)	(0.000348)	(0.000369)	(0.000408)	(0.000336)	(0.000348)	(0.000369)	(0.000408)
Analysts	0.0113**	0.0105**	0.00666	0.0116*	0.00977**	0.00877*	0.00503	0.0105*
	(0.00470)	(0.00506)	(0.00550)	(0.00607)	(0.00469)	(0.00506)	(0.00550)	(0.00606)
InstInve_VR	-0.00241	-0.00379**	-0.00464***	-0.00335*	-0.00252	-0.00394**	-0.00475***	-0.00338*
	(0.00159)	(0.00168)	(0.00178)	(0.00192)	(0.00159)	(0.00169)	(0.00178)	(0.00192)
DStkEng	1.222***	1.030***	0.822***	0.744***	1.246***	1.055***	0.853***	0.780***
	(0.0569)	(0.0633)	(0.0715)	(0.0812)	(0.0568)	(0.0632)	(0.0714)	(0.0811)
BoardSize	-0.00690	0.00510	0.00714	0.00264	-0.0164	-0.00511	-0.00519	-0.0137
	(0.0108)	(0.0118)	(0.0128)	(0.0142)	(0.0109)	(0.0119)	(0.0129)	(0.0143)
BoardActivity	-0.0104**	-0.0244***	-0.0282***	-0.0255***	-0.00962**	-0.0242***	-0.0281***	-0.0253***
	(0.00485)	(0.00577)	(0.00636)	(0.00736)	(0.00484)	(0.00577)	(0.00635)	(0.00736)
CEOduality	0.0787	0.160***	0.219***	0.322***	0.0694	0.150**	0.209***	0.309***
	(0.0559)	(0.0604)	(0.0660)	(0.0728)	(0.0560)	(0.0605)	(0.0660)	(0.0728)
BoardIndep	0.00577***	0.00576***	0.00591***	0.00610***	0.00312***	0.00301***	0.00269***	0.00310***
p. Im	(0.000966)	(0.00105)	(0.00115)	(0.00129)	(0.000654)	(0.000715)	(0.000793)	(0.000891)
BoardTenure	0.0151	0.0113	0.00562	0.00437	0.0121	0.00879	0.00316	0.00162
	(0.00977)	(0.0107)	(0.0118)	(0.0131)	(0.00976)	(0.0107)	(0.0118)	(0.0131)
CSRCommittee	1.686***	1.154***	0.809***	0.503***	1.676***	1.146***	0.796***	0.485***
	(0.0599)	(0.0664)	(0.0743)	(0.0832)	(0.0599)	(0.0664)	(0.0744)	(0.0833)
ERRI	-0.950***	-1.060***	-1.093***	-1.136***	-0.941***	-1.047***	-1.080***	-1.126***
	(0.132)	(0.140)	(0.147)	(0.157)	(0.133)	(0.140)	(0.147)	(0.157)

(continued on next page)

Fable 4 (Continued)

				Panel A. Female Directors typology	irectors typology	٨		
	Model 2 (t)	Lagged Model 2 (t-1)	Lagged Model 2 (t-2)	Lagged Model 2 (t-3) Model 3 (t)	Model 3 (t)	Lagged Model 3 (t-1)	Lagged Model 3 (t-2)	Lagged Model 3 (t-3)
EPI	0.0958***	0.0977***		***8060'0	0.0955***			0.0914***
	(0.0111)	(0.0117)		(0.0131)	(0.0112)			(0.0131)
Ω	0.808***	0.847***		0.849***	0.890***			0.955***
	(0.204)	(0.215)		(0.244)	(0.204)			(0.244)
Covid	1.821***	1.932***		1.919***	1.843***			1.948***
	(0.0580)	(0.0586)	(0.0605)	(0.0633)	(0.0578)	(0.0584)	(0.0603)	(0.0631)
industry	yes	yes		yes	yes			yes
Country	yes	yes	yes	yes	yes	yes		yes
Year	yes	yes	yes	yes	yes	yes		yes
Log Likelihood	-21,783.23	-19,513.59	-17,085.25	-14,549.81	-21,788.37	-19,523.06	-17,096.58	-14,557.76
p-value	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000

Additionally, in the first two columns it can be seen that there is a non-linear relationship (inverted U-shaped) between board gender diversity and climate change innovation, as it is confirmed that $6_1 > 0$ y $6_2 < 0$. In this respect, to determine the ideal number or proportion of female board members, we calculate the point $\Phi 1$ at which the change in the effect of these variables occurs, which is determined by $\Phi 1 = -(6_1 / 2 6_2)$. We found that this point is associated with the presence of 5 female directors on the board, which means 46.78% of the board members. Thus, when the number of female directors on the board surpasses five, climate change innovation does not correlate with the percentage of female directors or the statistical significance of the association between them decreases.

Furthermore, the relationships observed for the Inno_ClimaCh variable are confirmed in the models estimated for EnvScore and ESGScore, synthesized in panels B and C in Table 5. Therefore, it can be stated that the U-inverse relationship associated with board gender diversity is generalizable to any corporate sustainability decision made by boards of directors.

Discussion

Summary of findings

Our findings show that climate change innovation is positively associated with both the percentage and the number of female board members, indicating that, as posited by hypothesis H1, board gender diversity is associated with greater climate change innovation. Thus, the results support the theoretical predictions that female board representation favours the inclusion of climate change issues into corporate strategies and the adoption of proactive environmental initiatives in this regard, and are consistent with previous empirical evidence documenting a positive association between board gender diversity and eco-innovation (Galia et al., 2015; Glass et al., 2016; He & Jiang, 2019; Hobach & Jacob, 2018; Liao et al., 2019; Moreno-Ureba et al., 2022; Nadeem et al., 2020; Naveed et al., 2022) and climate change mitigation initiatives (Atif et al., 2021; García Martín & Herrero, 2020; Gull et al., 2022; Haque, 2017; Konadu et al., 2022; Nube & Velte, 2020).

However, this positive effect does not depend on the type of board female member considered (independent or executive director). In our sample, both independent and executive female directors have a positive and significant impact on climate change innovation, the impact being more pronounced for executive female directors than for independent female directors. Therefore, contrary to hypothesis H2, the positive effect of board gender diversity on climate change innovation is not due to the presence of independent female directors. This result is inconsistent with theoretical predictions that independent directors increase board effectiveness through better oversight, expertise and stakeholder orientation. Moreover, it contradicts previous empirical evidence (e.g., Atif et al., 2020, 2021; Gull et al., 2022), indicating a greater impact of independent female directors on environmental proactivity compared to executive female directors. This could be due to the fact that in our case the proportion of independent and executive female directors is relatively balanced, compared to other studies. In our view, a possible explanation for our result could be that, not belonging to the "old-boys' network", female directors tend to exhibit greater independence than their male counterparts (Adams & Ferreira, 2009; Li & Li, 2020; Seebeck & Vetter, 2022), which makes them better monitors regardless of their ties to the company, as independent or executive directors.

Our results partially support hypothesis H3. In line with its postulation, environmental proactivity in the fight against climate change increases as the number of female directors increases. Thus, the statistical significance of female board representation is greater for companies with three female directors than for companies with two, and is greater when there are two female directors on the board than

Table 5 Complementary analysis (*** p < 0.01, ** p < 0.05, * p < 0.1).

	Panel A. Inno_CliCh	1		P	anel B. EnvScor	re	P	anel C. ESGScor	e
	coeff	coeff	coeff	coeff	coeff	coeff	coeff	coeff	coeff
BoardDiversity(BD)	(std.error) 0.0494*** (0.00510)	(std.error)	(std.error)	(std.error) 0.241*** (0.0210)	(std.error)	(std.error)	(std.error) 0.362*** (0.0155)	(std.error)	(std.error)
Square_BD	-0.000528*** (0.000111)			-0.00175*** (0.000451)			-0.00288*** (0.000332)		
WomenDirectors (WD)	0.528***	(0.0438)		2.782***	(0.185)		3.807***	(0.136)	
Square_WD		-0.0518*** (0.00730)			-0.225*** (0.0305)			-0.295*** (0.0225)	
One_WD		(0.00750)	0.531*** (0.0617)		(0.0300)	2.338*** (0.260)		(0.0220)	4.009*** (0.192)
Two_WD			0.876*** (0.0797)			4.452*** (0.335)			6.263*** (0.249)
Three&More_WD		0.997***	, ,		5.442***			7.626***	, ,
ogAGE	0.863***	0.859***	(0.0914) 0.887***	5.953***	5.946***	(0.385) 6.201***	4.637***	4.667***	(0.285) 5.038***
logTA	(0.0571) 0.969***	(0.0570) 0.961***	(0.0570) 0.973***	(0.258) 6.284***	(0.258) 6.232***	(0.258) 6.322***	(0.192) 4.121***	(0.192) 4.061***	(0.195) 4.198***
ГobinQ	(0.0397) -6.989	(0.0397) -8.135	(0.0397) -4.446	(0.168) 292.2***	(0.168) 285.4***	(0.168) 306.6***	(0.123) 263.9***	(0.123) 256.9***	(0.124) 289.6***
_	(16.31)	(16.29)	(16.29)	(63.79)	(63.79)	(63.86)	(46.81)	(46.85)	(47.16)
ROA	-0.000352 (0.000588)	-0.000356 (0.000588)	-0.000351 (0.000588)	0.00163 (0.00264)	0.00162 (0.00264)	0.00163 (0.00264)	0.000949 (0.00194)	0.000899 (0.00194)	0.000975 (0.00195)
InterSales	0.000 (0.000)	0.000 (0.000)	0.000 (0.000)	0.000 (1.07e-10)	0.000 (1.07e-10)	0.000 (1.07e-10)	0.000 (7.87e-11)	0.000 (7.87e-11)	5.14e-11 (7.93e-11
R&D	2.64e-10	2.49e-10 (2.62e-10)	2.67e-10	3.88e-10 (1.08e-09)	3.10e-10 (1.08e-09)	4.41e-10	0.000	-1.14e-10 (7.95e-10)	0.000
Advertising	(2.61e-10) -1.42e-10	-1.36e-10	(2.62e-10) -1.44e-10	−9.22e-11	-6.70e-11	(1.08e-09) -1.16e-10	(7.94e-10) -9.88e-11	-6.76e-11	(8.00e-10 -1.47e-10
CAPEX	(1.18e-10) 0.000	(1.18e-10) 0.000	(1.18e-10) 0.000	(4.98e-10) 3.40e-10	(4.97e-10) 3.89e-10	(4.98e-10) 3.23e-10	(3.65e-10) 5.44e-10**	(3.65e-10) 6.03e-10**	(3.68e-10 5.24e-10*
Dividend	(8.56e-11) 0.000	(8.55e-11) 0.000	(8.58e-11) 0.000	(3.68e-10) -5.20e-10	(3.68e-10) -5.03e-10	(3.68e-10) -5.47e-10	(2.70e-10) 1.28e-10	(2.70e-10) 1.52e-10	(2.72e-10 6.45e-11
	(1.34e-10)	(1.35e-10)	(1.35e-10)	(5.89e-10)	(5.89e-10)	(5.90e-10)	(4.33e-10)	(4.33e-10)	(4.36e-10
Leverage	-3.95e-05 (0.000348)	-3.63e-05 (0.000348)	-6.32e-05 (0.000348)	0.00348** (0.00164)	0.00353** (0.00164)	0.00334** (0.00165)	0.00131 (0.00120)	0.00132 (0.00120)	0.00110 (0.00121)
Analysts	0.00988* (0.00506)	0.00967* (0.00506)	0.00835* (0.00505)	0.0692*** (0.0221)	0.0691*** (0.0221)	0.0572*** (0.0221)	0.0608*** (0.0162)	0.0605*** (0.0162)	0.0427*** (0.0163)
InstInve_VR	-0.00378**	-0.00368**	-0.00390**	-0.0121	-0.0116	-0.0131	-0.0165***	-0.0160***	-0.0182*
DStkEng	(0.00169) 1.038***	(0.00169) 1.028***	(0.00169) 1.051***	(0.00797) 9.057***	(0.00797) 8.998***	(0.00799) 9.159***	(0.00577) 7.248***	(0.00578) 7.181***	(0.00583) 7.438***
BoardSize	(0.0633) 0.000860	(0.0634) -0.0273**	(0.0633) -0.0218*	(0.267) 0.0852*	(0.268) -0.0984*	(0.268) -0.0528	(0.196) -0.117***	(0.197) -0.372***	(0.198) -0.303***
	(0.0118)	(0.0120)	(0.0120)	(0.0501)	(0.0510)	(0.0508) -0.0595**	(0.0368)	(0.0374)	(0.0375)
BoardActivity	-0.0240*** (0.00577)	-0.0234*** (0.00577)	-0.0238*** (0.00577)	-0.0602** (0.0241)	-0.0571** (0.0241)	(0.0241)	-0.0894*** (0.0177)	-0.0857*** (0.0177)	-0.0886* (0.0178)
CEOduality	0.155** (0.0605)	0.151** (0.0605)	0.146** (0.0605)	0.824*** (0.254)	0.799*** (0.254)	0.791*** (0.255)	1.852*** (0.187)	1.830*** (0.187)	1.797*** (0.188)
BoardIndep	0.00289***	0.00284***	0.00304*** (0.000715)	0.00801*** (0.00299)	0.00777*** (0.00299)	0.00935***	0.0178*** (0.00220)	0.0177***	0.0196***
BoardTenure	(0.000715) 0.0106	(0.000715) 0.0103	0.00738	-0.0424	-0.0491	(0.00299) -0.0715	0.0664*	(0.00220) 0.0547	(0.00221) 0.0202
CSRCommittee	(0.0107) 1.141***	(0.0107) 1.138***	(0.0107) 1.145***	(0.0466) 8.814***	(0.0465) 8.782***	(0.0466) 8.840***	(0.0341) 5.635***	(0.0340) 5.608***	(0.0342) 5.655***
ERRI	(0.0664) -1.090***	(0.0664) -1.102***	(0.0664) -1.055***	(0.286) -2.745***	(0.285) -2.857***	(0.286) -2.478***	(0.209) -0.793	(0.209) -0.873*	(0.210) -0.388
	(0.140)	(0.140)	(0.140)	(0.688)	(0.689)	(0.689)	(0.497)	(0.497)	(0.501)
EPI	0.0994*** (0.0118)	0.101*** (0.0118)	0.0986*** (0.0118)	0.536*** (0.0581)	0.549*** (0.0582)	0.527*** (0.0583)	0.202*** (0.0419)	0.216*** (0.0420)	0.188*** (0.0424)
EU	0.885*** (0.215)	0.856***	0.925*** (0.215)	5.710*** -1.065	5.512*** -1.067	6.077*** -1.067	2.859*** (0.768)	2.605*** (0.770)	3.491*** (0.776)
Covid	1.932***	1.931***	1.964***	2.756***	2.773***	3.003***	2.036***	2.089***	2.421***
Industry	(0.0586) yes	(0.0585) yes	(0.0583) yes	(0.228) yes	(0.227) yes	(0.227) yes	(0.168) yes	(0.167) yes	(0.168) yes
Country	yes	yes	yes	yes	yes	yes	yes	yes	yes
Year Lag Libalibaad	yes	yes	yes	yes	yes	yes	yes	yes	yes
Log Likelihood p-value	-19,508.86 0.000	-19,500.93 0.000	-19,520.33 0.000	-88,916.51 0.000	-88,908.07 0.000	-88,969.42 0.000	-81,848.34 0.000	-81,859.92 0.000	-82,046. 0.000

when female board representation is limited to a single female director. Likewise, the effect is greater when female board representation accounts for between 20% and 40% of the board size than when female directors occupy less than 20% of the board seats. Nevertheless, the presence of a single woman on the board has a significant impact on climate change innovation, without the need for a critical mass. These findings are in line with those obtained by Naveed et al. (2022) but contradict the results obtained by Ben-Amar et al. (2017), He and Jiang (2019), Hollindale et al. (2019), Moreno-Ureba et al. (2022), Nube and Velte (2021) and Torchia et al. (2011),

Our findings also indicate that there is a non-linear (inverted U-shaped) relationship between board gender diversity and climate change innovation. Thus, boards of directors with a balanced presence of men and women have a positive but not statistically significant effect. In addition, we found that when there are five or more female board members, which means 46.78% of the board, climate change innovation does not correlate with the percentage of female directors or the statistical significance of the association between them decreases. These results are in line with those obtained by Birindelli et al. (2019), Naveed et al. (2022) and Nube and Velte (2021).

Research and practical implications

This study complements previous research on the association between board gender diversity and eco-innovation and climate change mitigation initiatives by focusing on those innovations aimed at combating climate change, thereby responding to recent calls to explore the role of board gender diversity in shaping business initiatives to fight against climate change (Liao et al., 2015; Le Loarne-Lemaire et al., 2021). We provide empirical evidence of "who" within companies promotes innovation aimed at mitigating climate change and "how" and "under what conditions" they do so. Additionally, we support the need for a multi-theoretical framework that combines different theoretical lenses to provide a better understanding of the effect of the female board representation on environmental proactivity and, specifically, on climate change innovation.

Our results show that female board representation has a positive impact on climate change innovation and that this impact does not require a critical mass of female directors, although it is stronger as the number (percentage) of female directors increases until the threshold of five female directors (46.78% of the board members) is reached, at which point the relationship loses strength, suggesting a non-linear (inverted U-shaped) relationship between board gender diversity and climate change innovation. We also show that the positive effect of female board representation is stronger for executive female directors than for independent female directors, indicating that female directors influence climate change innovation mainly through the "executive power channel", owing to their direct involvement in management (Atif et al., 2020, p, 4357), rather than through the monitoring and advisory roles that characterize independent directors.

As for the practical implications of the study's findings, they provide valuable reasons for promoting gender diversity on boards: it favours the inclusion of climate change issues into corporate strategies, which positively affects innovation in climate change mitigation. Therefore, companies, investors and policymakers concerned about climate change should encourage a gender-balanced board composition as well as a greater involvement of female directors in management. At the societal level, our results show that promoting the fair participation of women on the board of directors is not only a matter of social justice but can also contribute to protecting the planet.

Conclusions

The fight against climate change has become a societal challenge that requires the coordinated efforts of a variety of actors, including companies whose investments in climate change mitigation technologies have grown in recent years (PwC, 2021). This paper responds to recent calls to examine the role of board gender diversity in shaping corporate responses to climate change (Le Loarne-Lemaire et al., 2021; Liao et al., 2015) by analysing whether female board representation is associated with greater climate change innovation and whether its effect is determined by the presence of a greater number of independent female directors or the existence of a critical mass of female directors.

The results obtained for a panel data of 3928 companies over the period 2010–2020 (35,199 observations) confirm that companies with a greater female board representation (both in number and in percentage) are more proactive in terms of investments in innovation for climate change mitigation. However, this positive effect does not require a critical mass of female directors and diminishes when the percentage of board seats held by women is higher than 46.78%, a proportion associated with the presence of five or more female directors. Furthermore, female directors influence climate change innovation mainly through their involvement in management as executive directors, rather than through the monitoring and advisory roles that characterize independent directors. The theoretical and practical implications derived from this research are robust to different considerations and methodological approaches.

Notwithstanding the interest and usefulness of our study, it suffers from some limitations. First, the results may be subject to potential endogeneity concerns due to omitted variables that may affect the relationship between board gender diversity and climate change innovation. Second, the findings could be caused by reverse causality (i.e. the possibility that female directors have self-selected proactive companies in the fight against climate change). Finally, the results may be affected by changing drivers of investments in climate change mitigation innovations and technologies (e.g. environmental regulation).

All these limitations should be addressed in future research. In addition, future studies could examine the moderating impact of some factors that usually drive environmental innovation and climate change mitigation initiatives on female directors' influence in this regard. Lastly, future research could consider the impact of female directors' background and demographic features on their influence on climate change innovation.

Funding

This work was supported by Junta de Castilla y León y Fondo Europeo de Desarrollo Regional [Grant/Award No. CLU-2019–03 Unidad de Excelencia "Gestión Económica para la Sostenibilidad" (GECOS)]; Consejería de Educación, Junta de Castilla y León [Grant/Award Number: SA069G18]; Ministerio de Ciencia, Innovación y Universidades [Grant/Award Number: RTI2018–093,423–B–I00]; Xunta de Galicia [Grant/Award Number: 2020 GPC GI-2016]; and Foundation for Science and Technology - FCT [UIDB/04,043/2020].

Declaration of Competing Interest

None.

References

Adams, R. B., & Ferreira, D. (2009). Women in the boardroom and their impact on governance and performance. *Journal of Financial Economics*, 94(2), 291–309. doi:10.1016/j.jfineco.2008.10.007.

Aibar-Guzmán, B., García-Sánchez, I. M., Aibar-Guzmán, C., & Hussain, N. (2022). Sustainable product innovation in agri-food industry: Do ownership structure and capital structure matter? *Journal of Innovation & Knowledge*, 7,(1) 100160. doi:10.1016/j.jik.2021.100160.

Amorelli, M. F., & García-Sánchez, I. M. (2020). Critical mass of female directors, human capital, and stakeholder engagement by corporate social reporting. *Corporate Social*

- Responsibility and Environmental Management, 27(1), 204–221. doi:10.1002/
- Ararat, M., & Sayedy, B. (2019). Gender and climate change disclosure: An interdimensional policy approach. *Sustainability*, *11*(24), 7217. doi:10.3390/su11247217.
- Arayssi, M., Dah, M., & Jizi, M. (2016). Women on boards, sustainability reporting and firm performance. Sustainability Accounting. Management and Policy Journal, 7(3), 376–401. doi:10.1108/sampj-07-2015-0055.
- Atif, M., Alam, M. S., & Hossain, M. (2020). Firm sustainable investment: Are female directors greener? Business Strategy and the Environment, 29(8), 3449–3469. doi:10.1002/bse.2588.
- Atif, M., Hossain, M., Alam, M. S., & Goergen, M. (2021). Does board gender diversity affect renewable energy consumption? *Journal of Corporate Finance*, 66, 101665. doi:10.1016/j.jcorpfin.2020.101665.
- Baker, H. K., Pandey, N., Kumar, S., & Haldar, A. (2020). A bibliometric analysis of board diversity: Current status, development, and future research directions. *Journal of Business Research*, 108, 232–246. doi:10.1016/j.jbusres.2019.11.025.
- Ben-Amar, W., Chang, M., & McIlkenny, P. (2017). Board gender diversity and corporate response to sustainability initiatives: Evidence from the carbon disclosure project. *Journal of Business Ethics*, 142(2), 369–383. doi:10.1007/s10551-015-2759-1.
- Birindelli, G., Iannuzzi, A. P., & Savioli, M. (2019). The impact of women leaders on environmental performance: Evidence on gender diversity in banks. Corporate Social Responsibility and Environmental Management, 26(6), 1485–1499. doi:10.1002/csr1762
- Boulouta, I. (2013). Hidden connections: The link between board gender diversity and corporate social performance. *Journal of Business Ethics*, 113(2), 185–197. doi:10.1007/s10551-012-1293-7.
- Caby, J., Coron, C., & Ziane, Y. (2022). The effect of top management team gender diversity on climate change management: An international study. Sustainability, 14(2), 1032. doi:10.3390/su14021032.
- Carter, D. A., Simkins, B. J., & Simpson, W. G. (2003). Corporate governance, board diversity, and firm value. Financial Review, 38(1), 33–53. doi:10.1111/1540-6288.00034.
- Chen, J., Leung, W. S., & Evans, K. P. (2018). Female board representation, corporate innovation and firm performance. *Journal of Empirical Finance*, 48, 236–254. doi:10.1016/j.jempfin.2018.07.003.
- De Villiers, C., Naiker, V., & Van Staden, C. J. (2011). The effect of board characteristics on firm environmental performance. *Journal of Management*, 37(6), 1636–1663. doi:10.1177/0149206311411506.
- Eagly, A. H., & Johannesen-Schmidt, M. C (2001). The leadership styles of women and men. *Journal of Social Issues*, 57(4), 781–797. doi:10.1111/j0022-4537.00241.
- Fama, E. F., & Jensen, M. C. (1983). Separation of ownership and control. *Journal of Law and Economics*, 26(2), 301–325.
- Galia, F., & Zenou, E. (2012). Board composition and forms of innovation: Does diversity make a difference? *European Journal of International Management*, 6(6), 630–650.
- Galia, F., Zenou, E., & Ingham, M. (2015). Board composition and environmental innovation: Does gender diversity matter? *International Journal of Entrepreneurship and Small Business*, 24(1), 117–141.
- García Martín, C. J., & Herrero, B. (2020). Do board characteristics affect environmental performance? A study of EU firms. Corporate Social Responsibility and Environmental Management, 27(1), 74–94. doi:10.1002/csr.1775.
- García-Sánchez, I. M., Aibar-Guzmán, C., & Aibar-Guzmán, B. (2020). The effect of institutional ownership and ownership dispersion on eco-innovation. *Technological Forecasting and Social Change*, 158, 120173. doi:10.1016/j.techfore.2020.120173.
- García-Sánchez, I. M., Gallego-Álvarez, I., & Zafra-Gómez, J. L. (2021). Do independent, female and specialist directors promote eco-innovation and eco-design in agrifood firms? Business Strategy and the Environment, 30(2), 1136–1152. doi:10.1002/bse.2676
- García-Sánchez, I. M., Raimo, N., Uribe-Bohorquez, M. V., & Vitolla, F. (2022). Corporate reputation and stakeholder engagement: Do assurance quality and assurer attributes matter? *International Journal of Auditing*, 26(3), 388–403. doi:10.1111/ iiau.12287
- Griffin, D., Li, K., & Xu, T. (2021). Board gender diversity and corporate innovation: International evidence. *Journal of Financial and Quantitative Analysis*, 56(1), 123–154. doi:10.1017/S002210901900098X.
- Gull, A. A., Atif, M., & Hussain, N. (2022). Board gender composition and waste management: Cross-country evidence. The British Accounting Review. doi:10.1016/j.bar.2022.101097 101097.
- Hambrick, D. C. (2007). Upper echelons theory: An update. *Academy of Management Review*, 32(2), 334–343. doi:10.5465/amr.2007.24345254.
- Hambrick, D. C., & Mason, P. A. (1984). Upper echelons: The organization as a reflection of its top managers. Academy of Management Review, 9(2), 193–206. doi:10.5465/ amr.1984.4277628.
- Haque, F. (2017). The effects of board characteristics and sustainable compensation policy on carbon performance of UK firms. *The British Accounting Review*, 49(3), 347–364. doi:10.1016/j.bar.2017.01.001.
- Haque, F., & Jones, M. J. (2020). European firms' corporate biodiversity disclosures and board gender diversity from 2002 to 2016. The British Accounting Review, 52,(2) 100893. doi:10.1016/j.bar.2020.100893.
- He, X., & Jiang, S. (2019). Does gender diversity matter for green innovation? *Business Strategy and the Environment*, 28(7), 1341–1356. doi:10.1002/bse.2319.
- Hollindale, J., Kent, P., Routledge, J., & Chapple, L. (2019). Women on boards and green-house gas emission disclosures. Accounting & Finance, 59(1), 277–308. doi:10.1111/acfi.12258.
- Horbach, J., & Jacob, J. (2018). The relevance of personal characteristics and gender diversity for (eco-) innovation activities at the firm-level: Results from a linked

- employer–employee database in Germany. Business Strategy and the Environment, 27(7), 924–934. doi:10.1002/bse.2042.
- Intergovernmental Panel on Climate Change (IPCC). (2007). Climate change 2007: The physical science basis. Available at Https://www.ipcc.ch/site/assets/uploads/2018/02/ar4-wg1-frontmatter-1.pdf [Accessed on 10th December 2022].
- Issa, A., & Bensalem, N. (2022). Are gender-diverse boards eco-innovative? The mediating role of corporate social responsibility strategy. Corporate Social Responsibility and Environmental Management. doi:10.1002/csr.2385.
- Joecks, J., Pull, K., & Vetter, K. (2013). Gender diversity in the boardroom and firm performance: What exactly constitutes a "critical mass? *Journal of Business Ethics*, 118 (1), 61–72. doi:10.1007/s10551-012-1553-6.
- Kanter, R. M. (1977). Some effects of proportions on group life: Skewed sex ratios and responses to token women. American Journal of Sociology, 82(5), 965–990. doi:10.1086/226425.
- Kassinis, G., Panayiotou, A., Dimou, A., & Katsifaraki, G. (2016). Gender and environmental sustainability: A longitudinal analysis. Corporate Social Responsibility and Environmental Management, 23(6), 399–412. doi:10.1002/csr.1386.
- Khandelwal, C., Kumar, S., Madhavan, V., & Pandey, N. (2020). Do board characteristics impact corporate risk disclosures? The Indian experience. *Journal of Business Research*, 121, 103–111. doi:10.1016/j.jbusres.2020.08.004.
- Konadu, R., Ahinful, G. S., Boakye, D. J., & Elbardan, H. (2022). Board gender diversity, environmental innovation and corporate carbon emissions. *Technological Forecast-ing and Social Change*, 174, 121279. doi:10.1016/j.techfore.2021.121279.
- Konrad, A. M., Kramer, V., & Erkut, S. (2008). The impact of three or more women on corporate boards. *Organizational Dynamics*, 37(2), 145–164. doi:10.1016/j.orgdyn.2008.02.005.
- Kramer, V. W., Konrad, A. M., Erkut, S., & Hooper, M. J. (2006). Critical mass on corporate boards: Why three or more women enhance governance. (pp. 2–4). Wellesley, MA: Wellesley Centers for Women.
- Le Loarne-Lemaire, S., Bertrand, G., Razgallah, M., Maalaoui, A., & Kallmuenzer, A. (2021). Women in innovation processes as a solution to climate change: A systematic literature review and an agenda for future research. *Technological Forecasting and Social Change*, 164, 120440. doi:10.1016/j.techfore 2020.120440
- Li, X., & Li, Y. (2020). Female independent directors and financial irregularities in Chinese listed firms: From the perspective of audit committee chairpersons. *Finance Research Letters*, 32, 101320. doi:10.1016/j.frl.2019.101320.
- Liao, L., Luo, L., & Tang, Q. (2015). Gender diversity, board independence, environmental committee and greenhouse gas disclosure. *The British Accounting Review*, 47(4), 409–424. doi:10.1016/j.bar.2014.01.002.
- Liao, Z., Zhang, M., & Wang, X. (2019). Do female directors influence firms' environmental innovation? The moderating role of ownership type. *Corporate Social Responsibility and Environmental Management*, 26(1), 257–263. doi:10.1002/csr.1677.
- Liu, C. (2018). Are women greener? Corporate gender diversity and environmental violations. Journal of Corporate Finance, 52, 118–142. doi:10.1016/j.j.corp-fin.2019.08.004
- Lu, J., & Herremans, I. M. (2019). Board gender diversity and environmental performance: An industries perspective. Business Strategy and the Environment, 28(7), 1449–1464. doi:10.1002/bse.2326.
- McGuinness, P. B., Vieito, J. P., & Wang, M. (2017). The role of board gender and foreign ownership in the CSR performance of Chinese listed firms. *Journal of Corporate Finance*, 42, 75–99. doi:10.1016/j.jcorpfin.2016.11.001.
- Midavaine, J., Dolfsma, W., & Aalbers, R. (2016). Board diversity and R&D investment. Management Decision, 54(3), 558–569. doi:10.1108/MD-09-2014-0574.
- Moreno-Ureba, E., Bravo-Urquiza, F., & Reguera-Alvarado, N. (2022). An analysis of the influence of female directors on environmental innovation: When are women greener? *Journal of Cleaner Production*, 374, 133871. doi:10.1016/j.jcle-pro-2022_133871
- Nadeem, M., Bahadar, S., Gull, A. A., & Iqbal, U. (2020). Are women eco-friendly? Board gender diversity and environmental innovation. *Business Strategy and the Environ*ment, 29(8), 3146–3161. doi:10.1002/bse.2563.
- Naveed, K., Khalid, F., & Voinea, C. L. (2022). Board gender diversity and corporate green innovation: An industry-level institutional perspective. Corporate Social Responsibility and Environmental Management. doi:10.1002/csr.2386.
- Nielsen, S., & Huse, M. (2010). The contribution of women on boards of directors: Going beyond the surface. *Corporate Governance: An International Review*, 18(2), 136–148. doi:10.1111/j.1467-8683.2010.00784.x.
- Nuber, C., & Velte, P. (2021). Board gender diversity and carbon emissions: European evidence on curvilinear relationships and critical mass. *Business Strategy and the Environment*, 30(4), 1958–1992. doi:10.1002/bse.2727.
- Pan, X., Chen, X., Sinha, P., & Dong, N. (2020). Are firms with state ownership greener? An institutional complexity view. Business Strategy and the Environment, 29(1), 197–211. doi:10.1002/bse.2358.
- Pfeffer, J., & Salancik, G. (1978). The external control of organizations: A resource dependence perspective. New York: Harper and Row.
- Post, C., & Byron, K. (2015). Women on boards and firm financial performance: A metaanalysis. *Academy of Management Journal*, 58(5), 1546–1571. doi:10.5465/ ami.2013.0319.
- Prado-Lorenzo, J. M., & García-Sánchez, I. M. (2010). The role of the board of directors in disseminating relevant information on greenhouse gases. *Journal of Business Ethics*, 97(3), 391–424. doi:10.1007/s10551-010-0515-0.
- PwC. (2021). State of climate tech 2021: Scaling breakthroughs for net zeroPwC. Retrieved from Https://www.pwc.com/gx/en/services/sustainability/publications/state-of-cli mate-tech.html.

- Seebeck, A., & Vetter, J. (2022). Not just a gender numbers game: How board gender diversity affects corporate risk disclosure. *Journal of Business Ethics*, 177(2), 395– 420. doi:10.1007/s10551-020-04690-3.
- Shaukat, A., Qiu, Y., & Trojanowski, G. (2016). Board attributes, corporate social responsibility strategy, and corporate environmental and social performance. *Journal of Business Ethics*, 135(3), 569–585. doi:10.1007/s10551-014-2460-9.
- Shui, X., Zhang, M., Smart, P., & Ye, F. (2022). Sustainable corporate governance for environmental innovation: A configurational analysis on board capital, CEO power and ownership structure. *Journal of Business Research*, 149, 786–794. doi:10.1016/j. jbusres.2022.05.037.
- Terjesen, S., Sealy, R., & Singh, V. (2009). Women directors on corporate boards: A review and research agenda. *Corporate Governance: An International Review*, 17(3), 320–337. doi:10.1111/j.1467-8683.2009.00742.x.
- Tingbani, I., Chithambo, L., Tauringana, V., & Papanikolaou, N. (2020). Board gender diversity, environmental committee and greenhouse gas voluntary

- disclosures. Business Strategy and the Environment, 29(6), 2194–2210. doi:10.1002/bse.2495.
- Torchia, M., Calabrò, A., & Huse, M. (2011). Women directors on corporate boards: From tokenism to critical mass. *Journal of Business Ethics*, 102(2), 299–317. doi:10.1007/s10551-011-0815-z.
- United Nations Global Compact (UNGC). (2012). A framework for corporate action on biodiversity and ecosystem services. Switzerland: UN Global Compact and IIICN
- Wang, D., Li, S., & Sueyoshi, T. (2018). Determinants of climate change mitigation technology portfolio: An empirical study of major U.S. firms. *Journal of Cleaner Production*, 196, 202–215. doi:10.1016/j.jclepro.2018.06.025.
 Zaman, R., Jain, T., Samara, G., & Jamali, D. (2022). Corporate governance meets corpo-
- Zaman, R., Jain, T., Samara, G., & Jamali, D. (2022). Corporate governance meets corporate social responsibility: Mapping the interface. *Business & Society*, 61(3), 690–752. doi:10.1177/0007650320973415.